Author's response to reviews

Title: Factors associated with age at first sexual initiation among youths in Gamo Gofa zone, south west, Ethiopia: A cross sectional study

Authors:

Marelign Tilahun Malaju Mr. (markunda76@Gmail.com)
Gistane Ayele Asale Miss. (gistanea@yahoo.com)

Version: 5 Date: 3 May 2013

Author's response to reviews: see over
Dear Mr Malaju,

Good day!

Your manuscript has now been peer reviewed and the Associate Editor wishes to pass on the following comments:

I have gone through the manuscript.

My comments are

The paper will require editing of the English.
This is best done by others and not the authors.
For example the abstract has 'in this study' repeated in the background, methodology and results section. This is redundant and unnecessary

- The redundant and unnecessary word is now removed

The manuscript should be written in past tense especially the Methods section where terms and variables were defined.

- All terms and variables defined are now written in past tense

3. Typos should be corrected eg Page 10 first paragraph 7th line of the discussion.

- Now this is corrected as required

4. Don't subtitle the limitation; rather let it flow with rest of the discussion.

- Done as required

5. In reference 3 in the reference section it is unclear what Napcc is. Write in full.

- Now it is written in full